Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — FEBRUARY 23, 2004 Here is the final homework assignment, which should be completed by Wednesday, March 11, 2004. This assignment will not be graded. As usual, “F&B” refers to the course text by Fraleigh and Beauregard.

Exercises 14, 16 and 18 beginning on p. 358 of F&B Exercise 10 beginning on p. 417 of F&B. Additional Exercise 2a for Section V.1 Exercises 12, 14 and 16 beginning on p. 429 of F&B Exercises 2, 4, 6 and 12 beginning on p. 437 of F&B Additional Exercises 1–3 for Section V.3